Author’s response to reviews

Title: Birth ball for pregnant women in labour research protocol: a multi-centre randomised controlled trial

Authors:

May Pui Shan YEUNG (may.yeung@cuhk.edu.hk)
Katrina Wai Kay TSANG (cuhkfm@cuhk.edu.hk)
Benjamin Hon Kei YIP (benyip@cuhk.edu.hk)
Wing Hung TAM (tamwh@cuhk.edu.hk)
Wan Yim IP (ip2013@cuhk.edu.hk)
Florence Wai Lei HAU (hwl081@ha.org.hk)
Margaret Kit Wah WONG (kw387@ha.org.hk)
Judy Wai Ying NG (nwy563@ha.org.hk)
Sau Ha LIU (liush2@ha.org.hk)
Sophia Shu Wing CHAN (sophiachen@cuhk.edu.hk)
Chi Kin LAW (c.law1@griffith.edu.au)
Samuel Yeung Shan WONG (yeungshanwong@cuhk.edu.hk)

Version: 1 Date: 23 Feb 2018

Author’s response to reviews:

The article has not sent to reviewers yet.